Author's response to reviews

Title: Re-emerging syphilis: a detrended correspondence analysis of the behaviour of HIV positive and negative gay men

Authors:

C PHILIP Wheater (p.wheater@mmu.ac.uk)
Dr Penny A Cook (p.a.cook@livjm.ac.uk)
Peter Clark (p.clark@livjm.ac.uk)
Qutub Syed (barbara.jones@hpa.org.uk)
Mark A Bellis (m.a.bellis@livjm.ac.uk)

Version: 5 Date: 15 Oct 2003

The paper has been reformatted as requested. The reviewer did not suggest any further changes.